Craig Chambers CSE 401 Introduction to Compiler Construction

Readings and Assignment #3

By: Read:

M 4/10 Reminder: Section 4.4, pp. 186-195 M 4/17 Section 4.9

By: Do:

M 4/10 Reminder: Scanner project, Exercises 3.16abc, 3.17abc F 4/14 Extend the EBNF description of the PL/0 syntactic structure to include the new language features, and hand in your revised description. (You’ll be given my grammar that day, so don’t start implementing the AST or the parsing code until you get my grammar.)

M 4/17 Exercises 4.1abe, 4.2, 4.3ac, 4.4bcd (bug fix for part d: use “a|b*a”as the sentence), 4.11a